

## **Author's response to reviews**

**Title:** Effectiveness and Safety of generic version of abacavir/lamivudine and efavirenz in treatment naïve HIV-infected patients: a nonrandomized, open-label, phase IV study in Cali-Colombia, 2011-2012

### **Authors:**

Jaime Galindo (jaimegalindo@cls.org.co)

Pedro Amariles (pedro.amariles@udea.edu.co)

Hector Mueses-Marín (centroinvestigaciones@cls.org.co)

Jaime A Hincapié (jaimealejandro.h@gmail.com)

Sebastián González-Avenidaño (sebastiangonzalez@humax.com.co)

Ximena Galindo-Orrego (ximeg78@hotmail.com)

**Version: 4 Date:** 19 Sep 2016

### **Author's response to reviews:**

Medellin, September 19, 2016

Dr. Nicola Gianotti

BMC Infectious Diseases

Subject: Response to some editorial revisions that were missed previously, in the revised manuscript "Safety and effectiveness of generic version of abacavir/lamivudine and efavirenz in HIV/AIDS treatment naïve patients: a nonrandomized, open-label, phase IV study in Cali-Colombia, 2011-2012" (INFD-D-16-00081R3)

Dear editor.

By logging on to the system, we have resubmitted the reviewed article “Safety and effectiveness of generic version of abacavir/lamivudine and efavirenz in HIV/AIDS treatment naïve patients: a nonrandomized, open-label, phase IV study in Cali-Colombia, 2011-2012" (INFD-D-16-00081R3)”.

According we had reviewed and addressed some editorial revisions, so the manuscript contains a Declarations section before the references, with ALL of the following subsections:

1. List of abbreviations

A list of abbreviations had been included. Page 20.

2. Ethics approval and consent to participate.

Ethics approval and consent to participate were included. Pages 20 and 21.

3. Consent for publication

Due to we are not publishing identifying patient data, we write "Not applicable" in this section. Page 21.

4. Availability of data and materials

The availability of data and materials had been included. Page 21.

5. Competing interests

It was denoted on page 21.

6. Funding

Funding was included. Page 21.

## 7. Authors' contributions

It was denoted on pages 21 and 22.

## 8. Acknowledgements

It was denoted on page 22.

Thank you for your collaboration and contribution to improve the quality of our manuscript.

Prof. Pedro Amariles

University of Antioquia